Author's response to reviews

Title: An outbreak of Streptococcus equi subspecies zooepidemicus associated with consumption of fresh goat cheese

Authors:

Markku Kuusi (markku.kuusi@ktl.fi)
Elina Lahti (elina.lahti@eela.fi)
Anni Virolainen (anni.virolainen@ktl.fi)
Maija Hatakka (maija.hatakka@cec.eu.int)
Risto Vuento (risto.vuento@pshp.fi)
Leila Rantala (leila.rantala@eela.fi)
Jaana Vuopio-Varkila (jaana.vuopio@ktl.fi)
Eija Seuna (eija.seuna@eela.fi)
Matti Karppelin (matti.karppelin@pshp.fi)
Marjaana Hakkinen (marjaana.hakkinen@eela.fi)
Johanna Takkinen (johanna.takkinen@ecdc.eu.int)
Veera Gindonis (veera.gindonis@saunalahti.fi)
Kyosti Siponen (kyosti.siponen@elintarvikevirasto.fi)
Kaisa Huotari (kaisa.huotari@ktl.fi)

Version: 5 Date: 16 January 2006

Author's response to reviews: see over
Dear David Holmes,

Thank you for the opportunity to submit the revised version of our manuscript ‘An outbreak of *Streptococcus equi* subspecies *zooepidemicus* associated with consumption of fresh goat cheese’ to the *BMC Infectious Diseases* Journal.

We have now made the requested formatting changes to the article:
1. Title and abstract are identical in the manuscript and in the submission system.
2. The list of authors in the manuscript is written exactly as they are in the submission system.
3. Section headings and subheadings are written as required.
4. ‘Materials and methods’ is renamed ‘Methods’.
5. ‘Authors contributions’ is changed to ‘Authors’ contributions’.
6. Mentions to table 1 is replaced by ‘see additional file 1’.

Yours sincerely

Markku Kuusi, MD
Department of Infectious Disease
National Public Health Institute
Mannerheimintie 166, 00300
Phone: +358-9-47448935
Fax: +358-9-47448468
E-mail: markku.kuusi@ktl.fi